**Author’s response to reviews**

**Title:** Surface ECG interatrial block-guided treatment for stroke prevention: Rationale for an attractive hypothesis

**Authors:**

Antoni Bayés de Luna (mmselles@gmail.com)

Manuel Martinez-Selles (mmselles@secardiologia.es)

Antoni Bayés-Genís (manuel.martinezsellers@universidadeuropea.es)

Roberto Elosua (manuel10@ucm.es)

Adrian Baranchuk (manuel.martinezselles@salud.madrid.org)

**Version:** 2 **Date:** 26 Jul 2017

**Author’s response to reviews:**

All changes have been done